Author's response to reviews

Title: Dialysis Disequilibrium Syndrome: Brain death following hemodialysis for metabolic acidosis and acute renal failure - A case report.

Authors:

Sean M Bagshaw (smbagsha@ucalgary.ca)
Adam D Peets (adpeets@ucalgary.ca)
Morad Hameed (morad.hameed@calgaryhealthregion.ca)
Paul JE Boiteau (paul.boiteau@calgaryhealthregion.ca)
Kevin B Laupland (kevin.laupland@calgaryhealthregion.ca)
Christopher J Doig (cdoig@ucalgary.ca)

Version: 3 Date: 19 August 2004

Author's response to reviews: see over
Biomed Central Nephrology

August 18, 2004

Dear Editors:

Regarding: MS: 1865654290378350 - Dialysis Disequilibrium Syndrome: Brain death following hemodialysis for metabolic acidosis and acute renal failure - A case report.

Thank you for accepting this manuscript for publication with Biomed Central Nephrology.

We have revised the manuscript in accordance with the Editor’s comments that were provided in your letter dated August 18, 2004.

Yours sincerely,

Sean M. Bagshaw, MD
Adam D Peets MD
Morad Hameed MD MSc
Paul JE Boiteau MD
Kevin B Laupland MD MSc
Christopher J Doig MD MSc

Dept of Critical Care Medicine
Foothills Medical Center
ICU Administration, Rm EG23
1403 - 29th ST NW
Calgary, Alberta, T2N 2T9
Tel: [403] 944-1570
Fax: [403] 283-9994